Author’s response to reviews

Title: Magnitude and Associated Factors of wasting among Under Five Orphans in Dilla Town, Southern Ethiopia: 2018: a cross-sectional study

Authors:
Adane Tesfaye Anbese (Adanetesfaye2006@gmail.com)
Tadesse Alemu Zerfu (tadal2005@yahoo.com)
Andnet Tadesse Wete (andnetvision@gmail.com)

Version: 2 Date: 04 Feb 2019

Author’s response to reviews:

To Reginald Adjetey Annan, Ph.D.

Warm Greetings to you, Thank you so much for the marvelous comments you gave us; we agree and accept all the comments.

Introduction of the abstract rewritten, clearly indicating rationale of the study

Rephrasing unclear statements as per the comments is done
Proportion of severely wasted orphans is mentioned now

We stated interventions the findings of the study lead us to propose

Information’s which demand referencing are referenced according to the comments

Grammar and spelling issues hardly tried to correct and rewritten

Under result and discussion we rephrase unclear statements commented

Limitations indicated

Fig-1 raised under result

Ethical issue made clearer and detail